Author’s response to reviews

Title: Positive patient experiences in an Australian integrative oncology center

Authors:

Bonnie J Furzer Dr (bonnie.furzer@uwa.edu.au)
Anna S Petterson Dr (anna.petterson@solariscare.org.au)
Kemi E Wright Ms (kemi.wright@uwa.edu.au)
Karen E Wallman Dr (karen.wallman@uwa.edu.au)
Tim R Ackland Dr (tim.ackland@uwa.edu.au)
David JL Joske Dr (david.joske@health.wa.gov.au)

Version: 4  Date: 15 April 2014

Author’s response to reviews:

Dear Editor,

Thank you for the reviews and please find compulsory revisions as requested by the associate editor.

Kind regards

Dr Bonnie Furzer